**Author’s response to reviews**

**Title:** DNA methylation alterations of AXIN2 in serrated adenomas and colon carcinomas with microsatellite instability

**Authors:**

- Yuta Muto (yutamuto@jichi.ac.jp)
- Takafumi Maeda (takafumim@hotmail.com)
- Koichi Suzuki (ksuzbnhm@omiya.jichi.ac.jp)
- Takaharu Kato (tkato@jichi.ac.jp)
- Fumiaki Watanabe (fwatanabe210@yahoo.co.jp)
- Hidenori Kamiyama (kamiyama@omiya.jichi.ac.jp)
- Masaaki Saito (msaito@jichi.ac.jp)
- Kei Koizumi (koizumi@hmedc.or.jp)
- Yuichiro Miyaki (miyaki@hama-med.ac.jp)
- Fumio Konishi (dzd00740@nifty.ne.jp)
- Sergio Alonso (salonso@imppc.org)
- Manuel Perucho (mperucho@imppc.org)
- Toshiki Rikiyama (trikiyama@jichi.ac.jp)

**Version:** 4  
**Date:** 21 May 2014

**Author’s response to reviews:** see over
BMC Cancer Editorial Board

Dear editorial board,

Please find enclosed our revised manuscript entitled “DNA methylation alterations of AXIN2 in serrated adenomas and colon carcinomas with microsatellite instability”, accepted for publication as a research article in BMC Cancer.

As indicated by the handling editor, we have made the following corrections:

1. We have included the Competing Interests section after the Conclusions section.
2. We have included the Authors Contributions section after the Competing Interests section.
3. We have copy-edited the manuscript with the help of a native English speaking colleague.
4. We have included all the authors emails on the title page, following the authorship order.
5. We have renamed the Introduction section as Background.
6. We revised the list of references, now listing all named authors regardless of the number. The term 'et al.' is not used.

I trust that with this revision the paper is suitable for publication in BMC Cancer.

Thank you very much for the positive consideration of our paper.

Yours sincerely,

Manuel Perucho, Ph.D.

ICREA Research Professor
Director
Institute of Predictive and Personalized Medicine of Cancer (IMPPC)
Carretera de Can Ruti, Camí de les Escoles S/n
08916 Badalona, Barcelona, Spain
phone: 34935543060
email: mperucho@imppc.org

Adjunct Professor,
Tumor Initiation and Maintenance Program
Sanford-Burnham Medical Research Institute (SBMRI)
10901 North Torrey Pines Road
La Jolla, California 92037
Phone: 858 6463112
Fax: 858 6463190
email: mperucho@sanfordburnham.org